Author's response to reviews

Title: If [we] get the disease, she will see 'pepper... and we will call it quits": HIV/AIDS Stigma and Utilization of Voluntary Counseling and Testing in Nigeria.

Authors:

Clifford O Odimegwu (Clifford.Odimegwu@wits.ac.za)

Version: 2 Date: 11 October 2012

Author's response to reviews:

I have been able to address the issues raised in the email sent to you. One, the title page and author's ids are included. Abstract is formatted according to the Journal's format. Three the study used a mixed-method approach and hence it is not a qualitative study perse.